

Contents lists available at ScienceDirect

Energy Conversion and Management: X



journal homepage: www.sciencedirect.com/journal/energy-conversion-and-management-x

Low-carbon technologies and just energy transition: Prospects for electric vehicles

Alaize Dall-Orsoletta^a, Paula Ferreira^{a,*}, Géremi Gilson Dranka^{a,b}

^a Algoritmi Research Center, University of Minho, Guimarães, Portugal

^b Department of Electrical Engineering, Federal University of Technology – Paraná (UTFPR), Pato Branco, Brazil

ARTICLE INFO	A B S T R A C T		
Keywords: Energy justice Electric vehicles Li-ion batteries Lifecycle stages Mining activities Whole Systems perspective	The impacts of low-carbon technologies are spread across countries and lifecycle stages in ways that can compromise the achievement of an inclusive and equitable energy transition. Based on an exploratory review, this paper identifies the main activities of the electric vehicles (EVs) life cycle, where they occur, and potentially associated injustices. Through a whole systems approach, energy justice tenets are extended to the EV technology, highlighting how it might fail to fully support a low-carbon and just energy transition. Results provide insights into how EVs can contribute to flexibility justice through smart grids and vehicle-to-grid developments, cosmopolitan justice as a consequence of greenhouse gas (GHG) emissions and global resource depletion, and restorative justice through laws and standards that demand environmental restoration and social compensation over affected communities. However, reviewed documents indicate that efforts must be directed toward reducing distributional, procedural, and recognition injustices across the North-South divide, especially those related to mining activities in the resource extraction and processing stage. EVs upfront costs and charging infrastructure issues may also exclude poor and rural communities during its operational stage. Recommendations for future research include technical aspects such as battery composition and recycling, which will determine the overall impact of EVs on resource extraction and of life stages, and social aspects of EV-technology such as social		

quantification of social impacts of low-carbon technologies.

1. Introduction

Energy transitions are socio-technical processes that encompass more than technological innovations. At the same time that energy transitions profoundly impact society, the economy, and the environment, their pace and reach also depend on social dynamics, such as social acceptance of technologies. Nevertheless, technocentric perspectives have dominated transitions' discourses [1], and social aspects have been commonly overlooked or instrumentalized with socio-economic gaps widening globally. From this background, efforts to include justice and equity as the focus of energy research have emerged. By assessing social impacts on individuals and communities, policymakers can implement measures that contribute to a just energy transition (JET). At first, the concept of a "just transition" originated in the 1970 s within trade union movements in the United States (US) [2], but since then it has been used in debates around "green new deals", "circular economy" [3], and even countries' Nationally Determined Contributions (NDCs), e.g., Chile [4]. Focused on energy systems and its social implications, the concept has evolved to the term JET, which has been applied to energy transitions research. The broad goal of a JET is to find ways of reconciling environmental sustainability with social and economic development globally, with a particular focus on countries that have a history of exploitation, low socioeconomic indicators, and struggle over energy access, security, and affordability [5].

innovations that can promote its inclusiveness, the achievement of the Sustainable Development Goals, and the

Likewise, a just energy system would be "a global energy system that fairly disseminates both the benefits and costs of energy services and one that has representative and impartial energy decision-making" [6]. However, the benefits and burdens of an energy system go beyond its operational stage, i.e., energy supply and demand, and spillover to other life cycle stages. Particularly, the United Nations (UN) Climate Change Conference (COP26) held in Glasgow in 2021 highlighted the criticality of resource extraction and usage for sustainable development, as "minerals demand for renewable energy technologies is projected to skyrocket, particularly for battery metals being used in electric vehicles - and the associated

https://doi.org/10.1016/j.ecmx.2022.100271

Received 5 December 2021; Received in revised form 24 May 2022; Accepted 23 July 2022 Available online 1 August 2022

^{*} Corresponding author. *E-mail address:* paulaf@dps.uminho.pt (P. Ferreira).

^{2590-1745/© 2022} The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

environmental and human impacts are likely to rise steeply as well" [7]. Special mention was made to electric vehicles (EVs) in this statement because of their expected contribution to decarbonizing the transport sector [8] and resource requirements for EVs battery manufacturing [9]. According to the International Energy Agency (IEA), the number of EVs should increase from 11 million in 2020 to 350 million in 2030, reaching almost 2 billion vehicles in 2050, in addition to a fourfold increase in the number of electric two- and three-wheelers and 50 million electric buses [10]. Nevertheless, it is not only the largeness of the electrified fleet that dictates the demand for raw materials but also batteries' size and capacity. Light-duty EV batteries can have a specific energy or energy density from around 30 to 275 Wh/kg with capacities varying from approximately 10 to 200 kWh [11]. Comparatively, the battery capacity requirements for a cellphone (e.g., Apple iPhone 6, 6.9 Wh) can be ten thousand times smaller than for an EV (e.g., Tesla Model S 85D vehicle, 85 kWh) [12]. Accordingly, the average mass of batteries used in popular EV models in the US is around 350 kg [13].

Conversely to internal combustion engine vehicles (ICEs), EVs are powered by battery-stored electricity. They are among the most promising technologies to decarbonize road transportation, along with sustainable fuels (e.g., advanced biofuels and electrofuels produced from electricity and CO2 [14]) and advanced powertrain vehicles (e.g., hybrid and fuel cells [15]). As batteries are fundamentally composed of two electrodes in an electrolyte [16], several different materials qualify as candidates for battery manufacturing. Among the battery technologies available for electric mobility, we can mention lead-acid, nickel--cadmium, and lithium-ion batteries (Li-ion batteries), each with a variety of electrode chemistry options [16]. Nevertheless, Li-ion batteries remain one of the most efficient and reliable battery systems for EVs [16]. Still, increasing the autonomy of EVs through enhancing battery capacities or reducing cars' energy consumption (e.g., lightweight design [15] and energy requirements of auxiliary systems [17]) will continue to be an important research stream if the foresought EVs uptake is to be fulfilled.

However, this upsurge in transport electrification may disrupt EVs supply chain and worsen social and environmental issues where related activities take place. To compound the problem, EVs supply chain relates to mining activities in a politically unstable country, the Democratic Republic of Congo (DRC) [18], where cobalt is extracted for Li-ion batteries production, and environmentally sensitive areas such as the Salar de Atacama in Chile [19]. Drawing from extensive fieldwork, Sovacool et al. [20] focused on cobalt mining in the DRC and electronic waste in Ghana and perceived that "decarbonization enhances existing vulnerabilities in the Global South, contributing to environmental degradation, ethnic and gender discrimination, and child labour" [21]. It also highlighted that much of previously published research has focused on future innovations, such as EVs, "but overlooks or obscures downstream and upstream processes, such as mining or waste flows" [20]. Therefore, considering the entire lifecycle of technologies is essential "from the front end where metals and minerals are extracted, to the back end where waste streams reside" [20].

This systemic analysis is aligned with the so-called whole systems approach, in which downstream and upstream processes are considered when evaluating a particular project or technology chain [22]. In order to provide a theoretical basis for discussions on justice, a 'multi-scalar' or 'whole systems' energy justice (EJ) approach was employed herein to evaluate EVs contribution to a JET. EJ has been addressed from a conceptual perspective (see for example, the principled approach [23], whole systems approach [24], and tenet-based approach [25]) with some attempts to better understand how technologies can contribute to EJ (e.g., hydropower in Vietnam [23], solar and wind projects in Chile, India, Kenya, and Mexico [26], utility-scale wind power in Mexico [27], the afterlife of off-grid solar products in Sub Saharan Africa [28], energy service contracts, EVs, solar panels, and low-carbon heating [29]). For the association of EJ and technologies, Lacey-Barnacle et al. [30] provided a systematic review of theoretical frameworks and energy foci of academic works on energy justice in the developing world, concluding that the largest bulk of papers addresses renewables and only 4 % of secondary papers included in the review focus on EVs. Sovacool et al. [31] evaluated the energy injustices (distributional, procedural, recognition-based, and cosmopolitan) of four low-carbon transitions nuclear power in France, EVs in Norway, solar energy in Germany, and smart meters in Great Brain - without employing a whole systems perspective. Along the same lines, Sovacool et al. [32] evaluated the same four cases using a whole systems perspective to give a spatial scale (micro, meso, and macro) for the injustices' extent. However, to the best of the authors' knowledge, the linkage between EV technology and JET considering a whole systems approach in association with the distributional, procedural, recognition-based, cosmopolitan, restorative, and flexibility EJ tenets is yet not explored. By employing a whole systems thinking, our study attempts to highlight the distribution of injustices across the globe that follow the EVs life cycle, from the extraction of resources for Li-ion battery manufacturing to the end of life of EVs. We focus on battery-powered light-duty road vehicles thanks to the significance of this low-carbon technology to decarbonize the transport sector and the larger sales volume of battery EVs in comparison to plug-in hybrid EVs [33].

Specifically, this work has the following objectives: (i) identify what are the main activities of EVs' life cycle, where they take place, and what are potential injustices associated with these activities; (ii) conceptually link potential injustices to EJ tenets; (iii) identify how EV uptake challenges the achievement of a JET, and (iv) highlight research gaps to be further addressed. The remainder of this paper is structured as follows. Section 2 provides a theoretical background on EJ, while Section 3 identifies this work's scope concerning the lifecycle activities of EVs. Section 4 describes the research methods and highlights potential EVrelated injustices found in the exploratory review. The linkages between EVs lifecycle stages and EJ tenets, and the technology contribution or opposition to a JET are discussed in Section 5. Lastly, Section 6 summarizes the main findings, limitations, and recommendations for future research.

2. Theoretical background: Energy justice

Even though the concept of EJ was firstly used in academia as a defined concept only in 2013 [34], it shares the same philosophical foundation as environmental and climate justice movements that began decades earlier [24]. EJ "*aims to provide all individuals across all areas with safe, affordable and sustainable energy*", [35]. There are three main approaches in the literature regarding EJ: (1) the principled approach [23], (2) the whole systems approach [24], and (3) the tenet-based approach [25]. In the principled approach, Sovacool et al. [23] reframed energy decisions as justice and ethical concerns and developed a framework for decision-making based on eight core principles: availability, affordability, due process, transparency and accountability, sustainability, intra-generational equity, inter-generational equity, and responsibility. Nevertheless, the principled-approach is less used than the tenet-based [36] and most of its principles can be seen as aspects of the tenet-based approach [37], which are described in the next section.

2.1. Energy justice tenets

The tenet-based approach has three central tenets: distributional, procedural, and recognition justices [25]. A recent review proposed by Jenkins et al. [36] comprehensively reviewed the EJ literature topic and identified that these three tenets are the most commonly used within EJ frameworks. **Distributional justice** calls for an even distribution of environmental ills and benefits and associated responsibilities regardless of income, wealth, gender, and color [35]. It also relates to the physical distribution of burdens and advantages, which has been used as an argument in local activities against wind development [27] and hydropower [38] projects, for example. **Procedural justice** refers to

including all stakeholders, particularly misrecognized parties, in the decision-making process in a non-discriminatory way [39]. Markedly, Jenkins et al. [25] explicitly took into account three mechanisms for achieving procedural justice: local knowledge mobilization, greater information disclosure, and better institutional representation. **Recognition justice** is about people being rightfully represented and having equal political rights [25]. It is concerned with political and institutional oppression, disrespect of local culture, and misrecognition of communities' interests and visions for the future [40].

In addition to these three main components, we highlight cosmopolitan and restorative justices [34]. Cosmopolitan justice suggests that "the ultimate unit of concerns are human beings and persons, not communities or nation-states" [41]. Therefore, it argues that moral and justice concerns apply to every-one equally as people are all members of a global community [41]. In contrast, **Restorative justice** calls for the energy sector to take responsibility (moral and financial) for negative impacts on the environment and surrounding communities and work towards restoring the landscape [34]. Finally, in a recent work evaluating justice in solar development, Heffron et al. [42] pointed to a new tenet: Flexibility justice, which is concerned with ensuring a market open to existing and new players and a flexible energy system. Flexibility justice is based on the idea of a flexible energy transition, in which all sectors have potential contributions if appropriately coupled, e.g., vehicle-to-grid technologies and smart mobility in the transport sector [43].

2.2. Whole systems

The whole systems approach supports a cradle-to-grave perspective for identifying the real impacts of energy projects through interactions among system components [24]. Without this systemic view, the impacts of extracting rare metals for manufacturing energy technologies, or biofuel production on biodiversity and land use, for example, cannot be wholly understood and, therefore, mitigated [44]. In energy systems modelling, a 'whole systems' appraisal would also consider the fuel supply chain [45]. Among the works that have employed a whole systems approach to EJ, we already mentioned Sovacool et al.'s work [32]. which evaluated four European low-carbon transitions and identified several injustices classified according to their spatial scales. Also, Sovacool et al. [22] developed a cradle-to-grave framework to evaluate energy transitions, and Mejía-Montero et al. [27] evaluated a utilityscale wind power in Mexico, providing "clear evidence against the notion that RETs (renewable energy technologies) [...] are inherently good or sustainable" [27]. Fig. 1 presents the combination of the tenet-based and

the whole systems approaches to EJ that will be explored in this study for the case of EVs.

3. Scope definition: The lifecycle stages of EVs

In this section, we review battery-powered EVs' life cycle according to four lifecycle stages: (1) resource extraction and processing, (2) manufacturing, (3) distribution and operation, and (4) waste and disposal. Even though transportation of car components takes place between lifecycle stages, we are not able to consider it fully herein, thanks to the complexities of evaluating freighting of all EV components. Alternatively, we consider the distribution of finished vehicles in the distribution and operation phase, battery shipping in the manufacturing phase, and reverse logistics in the end-of-life stage.

3.1. Stage 1: Resource extraction and processing

The first lifecycle stage includes mining and refining activities. For EVs, first-stage impacts are mostly linked to the extraction and processing of raw materials for energy storage as batteries are a key component and high contributor to EVs' cost and environmental impacts [46]. As the Global EV Outlook asserts, "depending on its size and assuming a typical range of emissions from battery manufacturing and the global average carbon intensity of electricity in the use phase, [EV battery] accounts for 10-30 % of the total life-cycle emissions" of battery-powered EVs [47]. Even though there are different types of batteries that employ different negative and positive electrode materials (see [16]), we consider Li-ion batteries for their reliability and technical efficiency and the fact that lithium-cobalt-manganese oxide cathode formulations have prevailed in the battery industry [48]. Raw materials commonly needed for manufacturing Li-ion batteries are manganese, cobalt, lithium, copper, nickel, aluminum, titanium, and carbon [49], whereas iron, rare-earth metals, and silicon are essential for electric motors [50]. Cobalt and lithium, in particular, were labeled as critical raw materials by the European Commission [51] and the US Geological Survey [52] because of their economic importance and supply risk. Therefore, the extraction and processing of lithium and cobalt were considered the two main activities of the first stage of EVs' lifecycle because of these materials' critical supply [51] and potential environmental impacts of Liion batteries (e.g., global warming potential, resource depletion, ecological toxicity, and human health [53]).

Lithium can be mined from salt-lake brines, hard-rock pegmatitic lithium ores, or lithium-rich clays, but, even though its occurrence in nature is common, few countries have significant economic exploitation



Fig. 1. The tenet-based and the whole systems approaches to EJ.

as is the case of Argentina, Australia, US, Chile, Bolivia, and China [19]. In Europe, Portugal has received attention thanks to its lithium reserves with the development of extraction and processing activities for EVs manufacturing being considered in the country [54]. According to the US Geological Survey, "five mineral operations in Australia, two brine operations each in Argentina and Chile, and two brine and one mineral operation in China accounted for the majority of world lithium production", [52]. Each of these countries produced over 5 million tons of lithium in 2020 [52]. After being mined, lithium must be further processed in chemical plants to remove impurities and produce lithium carbonate, which is then used for battery manufacturing. On the amount of raw material required, "it takes 250 tons of the mineral ore spodumene when mined, or 750 tons of mineral-rich brine to produce one ton of lithium" [55]. According to the Association of Mining and Exploration Companies, "China dominates lithium processing, accounting for an estimated 89 percent of the world's lithium hydroxide. Chinese refineries produce lithium carbonate, lithium hydroxide and lithium chloride — the precursors to lithium-ion battery cathode materials" [56]. Accordingly, China has the largest lithium consumption globally [12]. For cobalt, commonly produced as a byproduct of copper or nickel mining, the DRC was responsible for almost 70 % of the global cobalt mine production [52]. Apart from the DRC, which mined 95 million tons of cobalt in 2020, only Australia and Russia mined over 5 million tons of cobalt in the same year. The refinement of cobalt is also concentrated in China, the largest consumer of cobalt, which is used in the rechargeable battery industry [52].

3.2. Stage 2: Manufacturing

Resource extraction and processing are followed by manufacturing components and assembly into the final product. For EVs' batteries, refined raw materials have to be combined into cathodes, anodes, electrolytes, and separators [47]. The vast majority of battery components manufacturers are located in China, Japan, and South Korea [57]. After components are manufactured, they have to be combined into battery cells, which takes place mainly in China, the US, Europe, South Korea, and Japan [57]. Nevertheless, battery cell manufacturing has been taken near EVs manufacturing and assembly plants in Europe and the US mainly because of geopolitical tensions, risks and costs associated with shipping, and the security of supply [57]. Lithium can become highly inflammable when not packed and shipped appropriately, requiring temperature control, product enclosing, and stability during battery transportation. Therefore, if current trends continue, China will have 140 battery factories, Europe around 30, and the US, ten factories by 2030 [32]. Additionally, most EV assembly plants are found in China, the US, Japan, Germany, France, and the UK [57]. As the demand for EVs grows, so does manufacturing capacity, and some of the largest automotive companies have built or planned gigafactories to meet future demand (see [58]).

3.3. Stage 3: Distribution and operation

For distribution and operation, we consider the activities related to the sales and distribution and also re-charging and operation. Among the 40 countries with the largest total imports (US\$) of battery EVs, plug-in hybrid EVs, and full hybrid models, including Hong Kong as separate from mainland China, 31 are located in the Global North [59]. The five largest importers are Belgium, Norway, China, Netherlands, and Germany [59]. Similarly, considering total battery, plug-in hybrid, and full hybrid EV exports (US\$), among the top 40 countries, 34 are in the Global North [59]. The five largest exporters are Germany, the United States, South Korea, Belgium, and Slovakia [59]. Nevertheless, as our focus consist of battery-powered EVs, the next section evaluates injustices and trade values related to vehicles with electric propulsion only and Li-ion batteries. Once bought by end-users, EVs enter their operational lifetime (around ten years [47]). The largest concentration of EVs is in China, Europe, and North America. In 2020, China had 3.5 million battery electric cars and 1.0 million plug-in hybrid EVs, Europe 1.8 and 1.4 million, the US 1.1 and 0.6 million, and the rest of the world around 0.4 and 0.3 million [33].

3.4. Stage 4: Waste and disposal

The final stage comprises the waste and disposal of technologies, i.e., recycling, re-use, and waste. Recycling and reusing are part of a circular economy proposal, in which resource consumption and waste are reduced through the optimized utilization of by-products and the establishment of closed loops [60]. EVs recycling concerns, first, the dismantling of different components. The vehicle (without battery) and its traction battery follow different recycling paths. While some materials are wasted in landfills, vehicles undergo a shredding and postshredding process for metal scrap [61]. For batteries, some commercial recycling processes are available such as pyro-metallurgy, hydrometallurgy, hybrid and bio-processes, each with its advantages and disadvantages (see [62]), aiming to recover metals and cathode materials. In this lifecycle stage, once again, Li-ion batteries have been considered the primary concern as batteries re-use and recycling can reduce waste flows and raw materials extraction [63]. In other words, "widespread battery recycling can create a more stable domestic source of materials for battery production, reduce the demand for raw materials, and minimize the risks of geopolitical disruptions of the supply chain" [64]. However, there are some challenges such as the low volumes of spent batteries [62], the hazards of shipping and dismantling batteries [62], the incomplete recovery of valuable metals in commercial processes and the low economic value of some recovered materials [55]. As it is estimated that less than 5 % of Li-ion batteries get recycled at their end of life [65], most of them still end up in common landfills worldwide. Currently, Asia seems to dominate the battery recycling market with Sony Corporation in Japan [62] and companies such as Green Eco Manufacturing Resource (GEM), Hunan Brunp Recycling Technology, and Wuzhou Huayou Cobalt New Material in China [57].

At the end of EVs operation, batteries are assumed to retain 80 % of their initial capacity [47], making them eligible for use in several applications such as urban e-mobility and renewable energy storage, both at residential and utility scales [46]. Giving batteries a second-life is the focus of some pilot projects of car manufacturing companies such as General Motors (GM), Renault, Nissan, and BMW, and it has taken place mainly in Europe and the US [66]. Fig. 2 depicts the main regions where EV's life cycle activities occur. Also, it illustrates what would be a simplified flow for the typical trades across continents, including raw materials, batteries, and EVs.

4. Identifying injustices across EVs lifecycle stages

4.1. Research methods

This work explored works indexed in Science Direct, Scopus, and Web of Knowledge databases to identify potential injustices associated with EVs lifecycle activities. Sources in English were considered and research from 2012 onward was included in this study. A combination of EV- and justice-related terms were used, resulting in works linked to energy justice frameworks, as well as environmental and social life cycle analysis of EVs and associated materials/components. Given the wide scope and geographical resolution of this work a systematic literature review of works covering all EVs lifecycle stages would not be practicable.

When linking potential injustices to EJ tenets, six energy justice tenets of the tenet-based approach, i.e., distributional, procedural, recognition, restorative, cosmopolitan, and flexibility justices, were considered. Even though relevant to EJ literature, this work did not consider the principled approach as most of the latter's principles can be seen as aspects of the tenet-based approach [37]. Intra and intergenerational equity, for instance, which are pillars of sustainable



Fig. 2. Simplified map of EV's life cycle activities. The arrows represent flow of (1) lithium and cobalt for processing, (2) Li-ion batteries, and (3) EVs.

development, can be framed within distributive and cosmopolitan justice [41]. The six EJ tenets were combined with the whole systems approach to be as comprehensive as possible when evaluating the relationship between EVs and a JET across lifecycle stages. The proposed approach can also be extended to analyze other technologies' lifecycles. In the following subsections, injustices gathered from the exploratory review are synthesized according to related lifecycle stages and performed activities.

4.2. EVs and resource requirements: The burdens of cobalt and lithium mining

Mining activities have been linked to environmental justice issues due to air, soil, and water pollution [67], effects on human health and local biodiversity [68], water consumption [69], energy intensity [68], and conflicts with local and indigenous people [69]. Increasing demand has advanced mining activities toward more remote and unmined regions, which happen to overlap or be in the proximities of global protected areas in some cases [70]. Mining activities at the Lithium Triangle, i.e., Chile, Argentina, and Bolivia, are an example of overlapping with nature conservation areas [71]. In particular, lithium mining and low-efficient evaporitic technologies [72] in Chile have been linked to the forced delocalization of natives [71] and the depletion of water resources [67]. In Bolivia, as an opportunity for developing the national mining industry, it has generated local biodiversity concerns [73], whereas in the US, attempts to increase local production have produced conflicts with local communities [74].

Conflicts between mining companies and indigenous people are not new in the lithium supply chain and have already been documented in the lithium triangle [69]. Locals also seem to be blind-sighted due to poor knowledge sharing and are traditionally excluded in the decisionmaking process, even though their lives can be drastically affected by new mining activities [71], as they raise local pollution and health issues, for example [67]. In Portugal, even though lithium mining for battery manufacturing is only in the prospection and research stage, the initiative has already faced opposition from local communities thanks to poor communication and lack of transparency from companies and the government's side [75] and the negative impacts associated with the activity, e.g., effects on ecotourism, local agriculture, water consumption and pollution, and environmental degradation [76].

Concerning cobalt extraction, the struggles faced by Congolese people due to cobalt mining have been well documented by Sovacool [18], which highlighted pollution and health issues, terrible work

conditions for artisanal miners, conflicts between small and large-scale miners, corruption and malfeasance, exploitation and unfair market practices, forced delocalization of native people, and child labor [20]. Once batteries with cobalt and nickel cathodes have the highest potential for environmental impacts, reducing their use and/or exposure to these materials has been recommended [53]. Some progress has been made in replacing cobalt, but lithium is expected to continue as a critical element in batteries for energy storage and e-mobility [77]. In 2018, China dominated the production of refined cobalt with a 63 % share being responsible for producing of 78,152 tonnes per year [78]. According to Picarra et al. [78], the Herfindahl-Hirschman Index (HHI) for cobalt refining indicates a problematic "monopolistic hold of the market from China", which increases the vulnerability of other countries due to refined cobalt dependencies. The Chinese hold on the cobalt supply chain after cobalt extraction from DRC is also highlighted in Gulley et al. [79]. Lithium extracted from brines can be used in end-markets, whereas lithium extracted from hard rocks has to be further processed before being used in Li-ion batteries [56]. Australia, as the largest producer of lithium from spodumenes, has most of its lithium shipped to China for further processing in lithium carbonate, lithium hydroxide, or lithium chloride [56]. China is also the world's largest consumer of lithium thanks to its developed energy storage industry [12]. However, when comparing the manufacturing of battery components in China and the US, Hao et al. [80] found that the Chinese manufacturing industry can emit more than twice as much GHG emissions compared to the American due to the electrical grid composition and the energy intensity of industrial activities. Nevertheless, in average, the factory construction cost and the skilled labor rate is lower in China than other countries. For instance, the factory construction cost is around US\$ 333/m² in China, US\$420/m² in Poland, and US\$656/m² in the US, whereas the skilled labor rate for these same countries is US\$5/h, US\$6/h, and US\$85/h, respectively [56]. However, in addition to environmental aspects, there have been no social impacts and injustices related to lithium and cobalt processing alone documented in the reviewed literature.

4.3. Jobs, health, and emissions: Impacts of an electrified automotive industry

As China dominates the Li-ion batteries supply and is expected to continue at the top of raw material processing and manufacturing of battery cell components and cell manufacturing for the next years [55], it is reasonable to assess the Chinese manufacturing company's work conditions potential injustices. According to Sovacool [81], low-wage

factory workers in countries such as China and Malaysia are the ones *"inhaling toxic fumes and exposed to hazardous materials"* and seeing *"their lives shortened and environments degraded so that climate mitigation options can be used or exported elsewhere"* [81]. Furthermore, attention must be paid to the effects of low-carbon technologies on labor and employment rates. There is a concern about job losses and employee retention in the automotive manufacturing industry because of the substitution of ICE vehicles for EVs [82]. As gigafactories are being planned and constructed in Europe and the US [57], they can create work opportunities and foster regional and national economic development. On the other hand, their construction activities may negatively impact the environment as a large area is required for their establishment, and their operation requires high water and energy availability, which has created conflicts with local communities [83].

Moreover, it is not only the number of jobs that must be considered but the different skillsets needed in an electrified powertrain industry and the locations where these skills will be needed. Among the findings of Günther et al. [84] when evaluating the effects of fleet electrification mainly in China and Germany, there is a positive trend in job results from a whole location perspective [84]. However, the majority of jobs would be created in Eastern Europe (+3.9 million) against a + 0.1 million in China, which can be considered unsatisfactory figures due to the country's population size and demand for labor [84]. According to the IEA, the development of more sustainable battery supply chains alone could create + 10 million jobs worldwide [47]. However, Onat et al. [85] underscored that, even though EVs can create more green jobs, ICE and hybrid vehicles have the largest long-term employment potential, as they demand labor force in the petroleum and fossil fuel distribution industries. This is also highlighted by Di Felice et al. [82], who, after comparing different employment projections, stated that less labor and more energy seem to be required in the electric powertrain industry.

Nevertheless, a growing demand for EVs does not impact jobs alone. In the same way that the fossil-fuel industry brings health risks to society (e.g., it is estimated that fossil-fuel-related emissions are responsible for about 65 % of the excess mortality rate related to air pollution [86]), EVs



Fig. 3. Trade values by country as a percentage of total. . Adapted from [90]

and Li-ion batteries manufacturing can also affect populations' health and must receive governments' attention [13]. As a still-evolving technology, only recently is that the health hazards of Li-ion batteries started to be assessed, e.g., Sironval et al. [87] on the respiratory hazard of Liion batteries particles. In China, Shen et al. [88] analyzed the presence of heavy metals, including cobalt, on individuals from local communities in the Chinese battery industrial capital and identified dust as the main vector, being young children the most affected ones. Concerning environmental impacts, EVs manufacturing has been linked to higher GHG emissions than the fossil-fuel counterpart, but again, this is subject to the country's electricity mix and the efficiency of industrial processes [80]. In this vein, Xiong et al. [89] evaluated the lifecycle of EVs based on China's energy mix, and concluded that vehicle body and battery manufacturing account for the largest amount of energy consumption and GHG emissions.

4.4. Potential consequences of EVs ownership

As the EVs and electric batteries exports-imports trade values show in Fig. 3, EVs and battery trades are highly concentrated in Europe, Asia, and North America, highlighting a distributional injustice. Moreover, apart from China and South Korea, only few countries in the Global South are among the largest importers and exporters of EVs and electric batteries. For EVs importation and exportation, Europe dominates the market, whereas for batteries, Asia figures as the largest exporter with China holding an expressive 30 % share of total trade values [90]. The five largest importers of battery-powered EVs are Germany (13.5%), UK (13.1 %), Norway (9.4 %), France (7.5 %), and Netherlands (7.5 %) [90]. On the other hand, main exporters are Germany (24.4 %), US (15.8 %), South Korea (11.6 %), Belgium (11.2 %), and France (5.8 %) [90]. For battery imports, these are the trade value shares amongst continents: Europe (43.7 %), Asia (32.0 %), North America (18.7 %), Africa (2.8 %), South America (1.6 %), and Oceania (1.4 %) [90]. Even though it can be perceived a more distributed demand for batteries across countries, imports in Asian countries are most likely linked to an internal market. Conversely, the five largest exporters of electric batteries are China (30.4 %), South Korea (11.3 %), Germany (8.3 %), Poland (7.7 %), and Japan (6.6 %) [90].

Summed to these evident geographical disparities, EVs also highlight socio-economic imbalances as their purchase is limited to consumers of higher social classes due to high upfront costs [66], which has been referred to as the 'elitism of EVs' [41]. The current high purchase price is due mainly to battery costs [91]. However, Liu et al. [91] highlighted that, compared to some ICE models, the initial cost of EVs can be recovered in five years after purchase. Yet, authors advised that the economic advantages of powertrains vary according to geographic specificities due to electricity prices and matrix [91]. Moreover, the operation and distribution of EVs directly affect jobs related to vehicles' maintenance and fuelling. Furthermore, despite EVs job creation potential, created jobs are "often in different locations, skill sets, and sectors than the jobs that will be lost as fossil fuel decline" [10]. Around the 5 million jobs that could be lost by 2030, many are well paid and in the proximities of fossil fuel resources, "meaning structural changes can cause shocks for communities with impacts that persist over time", [10].

Besides distributional inequalities and job losses due to EVs distribution and operation, the substitution of ICE vehicles may pour millions of environmentally non-regulated second-hand vehicles into developing and less developed countries [92]. According to the UN Environment Programme, the three largest exporters of used vehicles are the European Union (EU), Japan, and the US, with 54 %, 27 %, and 18 % of the 14 million used light-duty vehicles exported between 2015 and 2018, respectively [92]. About 70 % of exported vehicles go to developing countries, which in the majority have limited or no regulations on safety and emissions standards of imported vehicles [92]. In Africa, for example, used vehicles comprise 85 % of the fleet, and most African countries do not have used vehicle standards or age restrictions in place

[93]. For instance, Doumbia et al. [94] evaluated the fleet composition in Abidjan, Côte d'Ivoire during a field study and identified that more than 60 % of personal cars are older than 10 years, with 6 % surpassing 30 years on the road. In opposition to newer vehicles holders of higher Euro standards, older cars in the fleet have been linked to worse air quality [95]. Betancourt et al. [96], when evaluating the exposure to air pollution from urban buses in Bogotá, Colombia, identified that "average exposure levels inside Euro II and Euro III (older buses) was twice as large as those measure inside vehicles with stricter emission standards, Euro IV or superior". In their study, the authors considered concentrations of fine aerosol particles, equivalent black carbon, and carbon monoxide, concluding that a fleet renewal would significantly reduce population exposure to them [96].

Furthermore, EVs have been linked to the rural-urban divide during their operation due to range constraints and lack of infrastructure to recharge vehicles, which may exclude distant communities, contributing to the urban-rural divide [32]. There is also a concern about the marginalization of other forms of mobility such as cycling and public transport, due to the predominance of battery-powered private cars [41]. Additionally, in countries with fossil fuel-based electricity mixes (e.g., China and India), the utilization of EVs causes emissions to be offset from tailpipe to power plants [82]. Dillman et al. [97] evaluated the GHG intensities of electric grids across the European Economic Area and concluded that for some countries, "EVs could lead to greater life-cycle GHG emissions than a comparable diesel counterpart" [97]. On the other hand, according to Moro and Lonza [98], the use of EVs instead of gasoline-fueled in EU member states can avoid, on average, 60 % of GHG emissions. However, this reduction may not be evenly distributed across countries since, if a Well-to-Wheel methodology that considers the carbon intensity of the electricity sector as well as electricity importsexports is employed, GHG emissions might increase [98]. This emphasizes the need for holistic and cross-sectoral approaches that connect energy supply to transport, industrial, and residential sectors, increase the share of renewable energy in electricity matrices [91], and develop smart grids [99]. Such holistic approaches can also avoid negative effects on electricity prices due to increased peak-load [100] since electrifying end sectors increases the electricity demand and grid flexibility requirements.

In this regard, vehicle-to-grid (V2G) technologies arise as options that allow a bidirectional flow of electricity from EVs to smart grids [99]. V2G can provide unused energy back to the grid during high peak demand periods [99], for instance. In a study in Portugal, Nunes and Brito [101] showed that even a small share of EVs for load balancing can positively affect grid flexibility as a means to reduce natural gas usage and energy management systems and numerically modeled the contribution of EVs and batteries to balance peak and off-peak periods, identifying a positive contribution.

4.5. Used batteries: trouble-maker or proponent of a regulated circular economy?

Spent Li-ion batteries are considered hazardous due to their content of heavy metals and have four main disposal routes: recycling, landfilling, illegal disposal, and informal processing [103]. If not disposed properly, batteries can contaminate water, soil, and air through a series of pollution routes such as leaching and dissolution [103]. Generated pollution can bring harm to the environment and human health, accumulate in soils, plants and crops, being battery toxicity harmful to various trophic levels [103]. Accordingly, Sovacool et al. [20] pointed out that informal processing and inappropriate disposal of Li-ion batteries are likely to expose workers and local communities to hazardous materials. Given the hazardous nature of Li-ion batteries, fire accidents at recycling stations, transport vehicles, and landfills have been reported in connection to used Li-ion batteries, for instance [104]. These fire accidents have impacted the waste facilities and surrounding communities through injuries, service disruptions, and monetary losses [104].

Nevertheless, recycling and reusing activities can go beyond environmental and health aspects and can create millions of jobs thanks to the "job-intensive nature of the logistics and industrial facilities needed to realize domestic recycling" [47]. Nevertheless, for this industry to grow, recycling processes have to be further developed to avoid inefficient recycling processes and waste generation [19]. According to Qiao et al. [61], in a study evaluating the economic and environmental benefits of EV recycling in China, the total potential reduction of energy consumption and GHG emissions per technology cost is 241.3 MJ/US\$ and 36.3 kg CO2eq/US\$, respectively. Nonetheless, the reduction from recycling vehicles without batteries is larger than the reduction achieved with battery recycling, as the latter process is not completely developed and subject to large efficiency improvements [61]. To reduce the loss of circularity and take advantage of the opportunities that accompany battery recycling, laws and standards that enforce responsible and safe disposal of low-carbon technologies and promote sustainable supply chains must be in place, which is not the case for most developing and less developed countries [47]. The European Commission has recently taken action towards establishing rules for battery design, production, and disposal as means to enforce environmental and social sustainability [105]. However, there is no universal standard battery recycling [103], even though lithium and cobalt, in particular, can significantly contribute to the UN SDGs and a circular economy agenda [65].

According to what has been presented in this section (Section 4), Table 1 summarizes EVs lifecycle stages (LCS), associated activities, most representative countries where these activities take place, and potential injustices.

5. The linkage between EVs lifecycle and EJ tenets

As one of the aims of this paper is the framing of injustices according to the energy justice tenets, this section brings, for the first time in the literature, the linkages between EVs lifecycle activities and distributional, flexibility, procedural, recognition-based, cosmopolitan, and restorative justices. By combining the energy justice topic with the EV technology, we intend to shed light on the most critical downsides of the road transport electrification, which are spread across EVs lifecycles, and provide policymakers and researchers with targets for action to counteract the identified downsides and strive for mitigating injustices.

5.1. Distributional and flexibility justices

The concept of distributional justice is especially relevant when investigating globalized supply chains to ensure that benefits and ills are evenly distributed. As it can be linked to both physical (environmental) and social burdens, distributional injustices are present in all EVs lifecycle stages. For lithium and cobalt extraction, burdens are borne by a small number of countries. The extraction of lithium from brines in the Lithium Triangle, i.e., Chile, Bolivia, and Argentina, has been linked to the regional depletion of water resources and local biodiversity concerns [72]. In contrast, cobalt extraction from the DRC has been linked to local pollution and health issues [18]. Cobalt and lithium processing are concentrated in China [56], and trade values indicate that economic gains through value-adding manufacturing activities in the EV lifecycle are contained in large economies in Asia, Europe, and North America [90]. The Chinese hold on lithium and cobalt processing activities gives it an almost monopolistic power over Li-ion batteries supply chain [78]. However, battery manufacturing has been linked to negative effects on local populations health in China [88]. Moreover, the construction of gigafactories in Europe, in an attempt to localize manufacturing activities in the continent, raised concerns over local water, land, and energy use [83]. Given the location of lifecycle activities with the most injustices, we can say that EVs contribute to the North-South divide. On

Table 1

N	lost representative countries	; and pote	ntial injusti	ices of EVs l	ifecycle activities.
---	-------------------------------	------------	---------------	---------------	----------------------

LCS	Activity	Most representative countries	Potential injustices
Resource extraction and processing	Lithium mining and processing Cobalt mining and processing	Chile, Australia, China, and Argentina ¹ Democratic Republic of Congo (DRC) ²	Knowledge asymmetry among stakeholders [71]; forced delocalization of natives [69,71]; conflicts between mining companies and indigenous people [69]; depletion of water resources [67,71]; local pollution and health issues [67]; local pollution and health issues [18]; terrible work conditions [18]; child labor [20]; conflicts
			between small and large-scale miners [18]; corruption and malfeasance [18]; exploitation and unfair market practices [18]; forced delocalization of natives [18]; monopolistic hold of cobalt processing [78];
Manufacturing	Battery components manufacturing Battery cell manufacturing	China, Japan, and South Korea ³ China, US, Europe ⁴ , South Korea, and Japan	Low-wage factory workers (e.g., China) [81]; construction of gigafactories [57] and conflicts with local communities [83]: job losses in the
	EVs manufacturing and assembly	China, US, Japan, Germany, France, and the UK [57]	automotive manufacturing industry [82]; health bazards [88]:
Distribution and Operation	Sales and Distribution	Main importers vs Main exporters ⁵	Job losses in the fossil fuel industry [41]; EVs purchased by consumers with high socioeconomic conditions [41]; distributional inequality across countries due to uneven uptake [33,90]; price of EVs [66]; second-hand ICE markets [92];
	Re-charging and Operation	China, Europe, and the US ⁶	Urban-rural divide [32]; marginalization of other forms of mobility [41]; job losses in the fossil fuel industry [10]; offset of emissions from tailpipe to power plants [82]; North- South divide [20]; adverse effects on electricity price due to increased peak- load [100]:
End of life		Worldwide ⁷	Spent Li-ion batteries have been linked to (continued on next page)

A. Dall-Orsoletta et al.

Table 1 (continued)

LCS	Activity	Most representative countries	Potential injustices
	Waste and re-use of batteries (Second life) Recycling of batteries	China (>60 %) and South Korea [57]	pollution and health hazards [103], accidents and injuries [104], informal recycling and inappropriate disposal to environmental and health issues [20]; inefficient recycling processes and waste generation to resource depletion and loss of circularity [19];

¹ Countries with a lithium mine production of over 5 million tons in 2020 [52]. ² DRC is responsible for 70 % of global cobalt production in 2020 [52]. ³ China, Japan, and South Korea are home to a significant share of Li-ion batteries components manufacturing capacity: cathodes (85 % of global capacity), anodes (97 %), separators (84 %), and electrolytes (64 %) [106]. ⁴ Battery-cell manufacturing in Europe is concentrated mainly in the United Kingdom, France, and Germany [57]. ⁵ Main importers: Germany, UK, Norway, France, Netherlands, and Main exporters: Germany, US, South Korea, Belgium, France [90]. ⁶ EV stock in 2019: China 3.3 million, Europe 1.8 million, and the US 1.5 million [33]. ⁷ Battery second life projects: Germany, US, France, UK, Netherlands, Japan [66].

the discussion over job creation and losses, even though some works point to an overall positive net employment, there will be a possible mismatch between created jobs and the skills and the location of available workforce [82], which highlights a both positive and negative effect of EVs on employment.

Likewise, benefits due to emissions and air pollution reduction, which are expected in EVs' operational phase, are likely to materialize in urban environments in a selected number of countries in the Global North, depending on countries' electricity matrices [98]. China, for instance, even though concentrating the largest number of EVs, has seen an increase in emissions from the transport sector as 65 % of its electricity was generated from coal in 2019 [53]. This offset of emissions from tailpipes to powerplants [107] stresses the importance of integrating renewable energy into energy mixes and having flexible demand strategies. Nevertheless, there is a clear advantage when offsetting tailpipe emissions from power stations since it is simpler to manage and reduce emissions from a few sources than from countless ICE vehicles on the road [108]. Besides, an asymmetrical EV uptake across continents has consequences for the market of used ICE vehicles, commonly directed to developing countries in Africa and South America and linked to air pollution and safety, energy efficiency, and operational costs issues [92]. On the other hand, a second-hand EV market can help less developed countries accelerate their fleet electrification if electric cars reach domestic markets with accessible purchasing prices [92] since another aspect that contributes to the distributional injustice of EV technology is its high upfront costs, which limits its acquisition to wealthier socio-economic groups. Nevertheless, costs are expected to considerably fall with technological development [109].

Lastly, good management of used EVs at the end-of-life stage can help establish sustainable supply chains through reuse and recycling loops of batteries. This is a determinant aspect of Li-ion batteries impact so they do not end up in common landfills or are dismantled by workers under unsafe and unhealthy conditions, affecting local health and biodiversity, as already reported in the literature [20]. Batteries' recycling is currently dominated by Asia and is subject to efficiency improvements before it becomes financially and environmentally attractive [61]. Flexible production and waste management processes are also linked to a flexible energy transition [43] in conjunction with reusing batteries for energy storage systems. The flexibility justice tenet relates to EVs thanks to sector coupling and its effect on energy demand and storage.

Concerning costs for re-charging at the operational phase, appropriate market pricing mechanisms and strategic charging can help avoid large-scale EVs charging at times of peak demand, which would impact grid stability and energy price. Higher electricity prices have been linked to energy poverty issues as raising prices inflict an extra burden on a household that do not have large disposable incomes [110]. V2G technologies can also help reduce costs of infrastructure changes [111] and contribute to a flexible energy transition by balancing variable energy supply from RES and demand [43]. V2G services depend on several technical aspects, but it could "unlock up to 600 GW of flexible capacity distributed across the main electric vehicle markets in 2030 and moderate intermittency of variable renewables during peak demand" [47]. Fig. 4 represents the main aspects to be considered concerning distributional justice and flexibility justice according to the lifecycle stages together with a positive (+), negative (-), or combined sign (-+) indicating the potential effect of EVs. As it can be seen, distributional injustices, which indicate a potential negative effect of EVs and offer opposition to a JET, are fairly distributed across lifecycle stages. On the other hand, EVs seem to have the potential to positively affect flexibility justice mostly during the distribution and operation phase. Accordingly, Li-ion batteries recycling and reuse rate will dictate whether the technology contributes to or opposes a JET concerning natural resources usage at resource extraction and waste and disposal lifecycle stages.

5.2. Procedural and recognition justices

Knowledge asymmetry and stakeholder conflicts are particularly relevant for justice issues at the resource extraction and processing stage. Increasing demand has advanced mining activities toward more remote and unmined regions, which happen to overlap or be in the proximities of global protected areas in some cases [70], as in the Lithium Triangle [71]. Even though Chile ratified the ILO Indigenous and Tribal Peoples Conventions in 2008, internal factors and governmental policies have limited the participation of indigenous people in decision-making processes about land requirements and water use, once their participation can harm the mining industry interests and intensify conflicts between indigenous people and the mining industry [69]. This is associated with the injustices presented in the previous section related to the forced delocalization of native people and the abandonment of ancestral settlements due to water scarcity as the mining industry requires large amounts of water to operate in one of the driest regions in the world [67,71]. As ensuring lithium supply becomes one of the priorities of carmakers, countries want to become independent in both sourcing and manufacturing EV technology. However, this rush to extract more lithium can lead to more conflicts, environmental degradation, and disregard for local communities' interests. Even at the European level, conflicts about the potential expansion of lithium exploitation activities are emerging, as mentioned in the previous section for the case of Portugal, where mining locations pose challenges to local communities' wellbeing and functioning [76]. Concerning cobalt mining, besides critical environmental and human health issues [68], mining activities have been associated with the displacement of indigenous communities, corruption, malfeasance, political oppression, and conflicts in the DRC [18]. Moreover, the construction of gigafactories for battery manufacturing may also generate conflicts with local communities and produce procedural injustices [83]. And not only gigafactories, but industrial practices that negatively affect the health and safety of surrounding populations often lack the participatory processes and decision-making that should be in place concerning people's interests. Fig. 5 summarizes the linkages between EVs lifecycle stages and procedural and recognition justices. As it can be seen, activities at resource extraction and processing and manufacturing stages have been



Fig. 4. Distributional and flexibility justices: Potential effects of EVs.



Fig. 5. Procedural and recognition-based justices: Potential effects of EVs.

offering opposition to establishing a JET.

5.3. Cosmopolitan and restorative justices

Concerning cosmopolitan justice, EVs have been related to the consumption of finite mineral resources [9], toxicity increase [48], water scarcity [76], and a shift from tailpipe to power plant emissions in cases where electricity supply is heavily dependent on fossil fuels [85]. These issues inflict a burden on the environment as a whole and have intra and intergenerational consequences through GHG emissions, natural resource depletion, and climate change. In case EVs fail to reduce overall GHG emissions, this technological choice will not help the world stay below the 1.5 °C global average temperature threshold, which will bring serious and damaging effects to the current and future generations. These aspects must compel us to act as global citizens towards building and inclusive and resource efficient future [112], in the same way they prompt interlinked and holistic sustainability assessments for intergenerational 'strong sustainability' [113]. In circumstances in which social and environmental damage is unavoidable, such as mining activities, restorative justice must compel companies to take responsibility for optimizing processes as much as possible, maintaining sustainable practices, and compensating the most impacted people by improving local working conditions, respecting local cultures and interests, as well as access to essential services in order to avoid 'green colonialism' practices [114]. This is clearly not the case in many mining sites around the world, where the environment is not appropriately restored and surrounding communities are not given the means to thrive after the end of mining activities (e.g., Annandale et al. [115] on the impacts of mining on Indigenous livelihood and the importance of pre- and postmining management).

Restorative justice may also imply establishing specific laws and standards according to the activities developed in each region, such as mandatory recollection of used batteries by manufacturers through reverse logistics to avoid dumping in inappropriate fields [47]. These laws and standards are expected to establish socially and environmentally sustainable supply chains, as referred to in EU regulations over batteries sourcing and recycling [105]. Therefore, restorative justice obliges companies to take moral and financial responsibility for their operations when they negatively impact society, as well as biding to laws and standards, restoring the environment in cases of degradation, decurrent pollution, and accidents, and social compensation through the improvement of essential local services and job opportunities, for instance.

Fig. 6 encapsulates the aforementioned relationships. Within restorative justice, negative effects on social compensation and environmental restoration are contained to the resource extraction and processing stage, while we attribute to laws and standards a potentially positive effect on the sustainability of the entire EVs supply chain. Nevertheless, similar to V2G technologies, this potential positive effect is yet to materialize. As expected, aspects linked to cosmopolitan justice are evenly spread across lifecycle stages. Particularly, pollution and effects on biodiversity receive a potential positive effect in addition to the negative as, in the long term, the substitution of ICE vehicles by EVs can reduce the carbon dioxide concentration in the atmosphere and, therefore, climate change effects.

6. Conclusion

A whole systems perspective over low-carbon technologies can contribute to a JET since, by analyzing what activities are performed in each lifecycle stage, where they take place, and which communities are affected, injustices can be identified and further addressed. However, there can be no one-size-fits-all solution, given each country's different challenges. Decarbonizing the transport sector is essential for reducing GHG emissions, and EVs are undoubtedly playing a key role in this process. However, our exploratory review shows that efforts must be directed toward reducing injustices that are spread across lifecycle stages and countries. Among the most critical aspects of this work, we highlight the distributional injustices caused by: (i) the disregard of social and environmental aspects in lithium and cobalt mining activities, (ii) work, emissions, and health-related issues during manufacturing stages, (iii) the concentration of value-adding activities in China and developed countries which can potentially widen cross-country and continent inequality, (iv) EVs infrastructure and charging requirements that can exclude rural communities, (v) EVs price which sustain the technology's elitism [41] (vi) the market of used ICE vehicles which can transfer emissions and health issues associated with ageing and less efficient fleet mostly to developing and less developed countries in South America and Africa, and (vii) the lack of recycling and reusing activities at waste and disposal waste of Li-ion batteries and its associated health and environmental hazards. Also, the extension of flexibility justice to EVs shows the relevance V2G technologies can have in

ensuring a more reliable and flexible energy system. The effects of EVs on this EJ tenet are mostly contained to the distribution and operational stage due to energy-transport coupling. Also, considering that a flexible energy system would also imply compliant resource management practices, EVs can positively or negatively contribute to a reduction in resource usage depending on recycling and reuse rates. Conversely, EVs can burden electric grids due to high peak demand and poor households in case demand and supply are ill-coordinated. This may cause electricity prices to go up, restricting the fight against energy poverty and the achievement of SDG7.

The framing of injustices according to procedural and recognitionbased tenets showed that the most critical stage for attaining a JET from this side is the resource extraction and processing lifecycle stage. Our review showed that lithium and cobalt extraction impacts on surrounding communities are commonly neglected, causing local conflicts, and the forced delocation of native people. Moreover, activities in the DRC have been linked to corruption and unethical work aspects [20]. In the manufacturing phase, injustices may also appear within industries and their vicinities as participatory decision-making is also missing. For sustainable and ethical supply chains, adequate international laws and standards are critical. These standards are directly connected to companies legal obligation to restore the environment and the communities affected by their activities. These are key aspects of restorative justice, which are currently and mostly required at the resource extraction stage. Lastly, the use of finite resources, global emissions, and climate change put a lot of pressure on EV-technology development and uptake as the ultimate goal of EVs is to reduce transport emissions and fossil fuel dependency. However, even though EV-related GHG emissions reductions can be contradictory in the literature, especially when a lifecycle perspective is employed, low-emission electricity matrices, technological development, and flexible energy systems can provide the technology with the means to fulfil its purpose. Concerning resource usage, EVs' positive or negative effects continue to depend on the enhancement of recycling processes, their regulation and distribution across countries, as well as the development of energy storage technologies.

Therefore, in relation to whether EVs contribute to or oppose a JET, our exploratory review shows that it depends. From one side, EVs have a great potential to reduce GHG emissions, create jobs in the electric powertrain and recycling industries, enhance energy systems flexibility, improve health in urban environments, and reduce energy and resource consumption. On the other hand, they can increase socio-economic inequalities across countries, perpetuate stories of green colonialism, stretch gaps between socioeconomic classes, add a load to natural resources extraction, negatively affect health and biodiversity through air,



Fig. 6. Cosmopolitan and restorative justices: Potential effects of EVs.

water, and soil pollution, and even increase GHG emissions across the whole lifecycle if renewable energy integration does not happen in tandem with EVs uptake. Nevertheless, even though we do not discuss the injustices associated with fossil fuel chains and ICE vehicles, we acknoledge they are many and widespread. As EVs are an essential part of transport decarbonization plans, they should have its most critical environmental and social aspects identified and addressed. Since lowcarbon technologies are not essentially just or inclusive, as our analysis of EVs shows, action must be taken to reduce inequalities and unintended negative impacts.

Among the limitations of this work, we could cite the exploratory nature of the review and the lack of quantitative analysis to support justice claims for the case of EVs. The extent to which these findings have wider relevance should be further evaluated. Moreover, our assessment of EVs lifecycle and associated injustices at resource extraction and waste and disposal was limited to Li-ion batteries, which makes us recognize the scope limitations of this work. Accordingly, this work was also limited to energy storage technologies employing lithium and cobalt. However, as a rapidly evolving technology, battery chemistry can reach a point where neither of these raw materials is required [11].

Given the interdisciplinarity of this work, we have future research recommendations that include technical and social aspects. In technical terms, as already highlighted by many works (e.g., [46,116]), it is necessary to keep working on (i) the development of batteries that use less critical raw materials, e.g., cobalt and lithium, without impairing EVs performance; (ii) the improvement, regulation, and dissemination of battery recycling processes; (iii) research on battery re-use and development of pilot projects and (iv) grid stability and flexibility and the effects of large-scale EV charging on electric grid and price. Additionally, the appropriateness of EVs and other modes of transport in specific contexts and regions, following advances made in other transport-related technologies (e.g., biofuels [117]) must be analyzed. Ergo, research on the impact of policies on emissions, socioeconomic aspects, and alternative fuel vehicles' uptake is critical [118]. On the social side, we recognize the relevance of (v) research on social innovations and new business models for overcoming socio-economic barriers imposed by EVs upfront and infrastructural costs and associated distributional inequalities. Likewise, (vi) research and funding towards ensuring access to clean and affordable energy for all, as proposed by SDG7, can reduce wealth inequalities and, in the long run, contribute to EVs uptake. (vii) As car manufacturers rush toward ensuring supply, the impacts of localized supply chains on environmental, social, and market dynamics both at international and national, must be further explored. (viii) Lastly, research on justice aspects of low-carbon technologies that employ methods to quantify aspects related to technologies' justice can substantially contribute to a JET by providing ways of measuring progress.

CRediT authorship contribution statement

Alaize Dall-Orsoletta: Conceptualization, Visualization, Writing – original draft, Writing – review & editing. Paula Ferreira: Conceptualization, Supervision, Visualization, Writing – review & editing. Géremi Gilson Dranka: Conceptualization, Visualization, Writing – review & editing.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgments

This work has been supported by FCT – Fundação para a Ciência e

Tecnologia within the R&D Units Project Scope: UIDB/00319/2020.

References

- [1] Labanca N, Pereira AG, Watson M, Krieger K, Padovan D, Watts L, et al. Transforming innovation for decarbonisation? Insights from combining complex systems and social practice perspectives. Energy Res Soc Sci 2020;65:101452. https://doi.org/10.1016/j.erss.2020.101452.
- [2] Stevis D, Felli R. Global labour unions and just transition to a green economy. Int Environ Agreements Polit Law Econ 2015;15:29–43. https://doi.org/10.1007/ s10784-014-9266-1.
- [3] Healy N, Barry J. Politicizing energy justice and energy system transitions: Fossil fuel divestment and a "just transition". Energy Policy 2017;108:451–9. https:// doi.org/10.1016/j.enpol.2017.06.014.
- [4] Gobierno de Chile. Chile's Nationally Determined Contribution. 2020.
- [5] García-García P, Carpintero Ó, Buendía L. Just energy transitions to low carbon economies: a review of the concept and its effects on labour and income. Energy Res Soc Sci 2020;70:101664. https://doi.org/10.1016/j.erss.2020.101664.
- [6] Sovacool BK, Dworkin MH. Global Energy Justice 2014. https://doi.org/ 10.4324/9781315762203.
- [7] Declaration on Mining and the Energy Transition for COP26. Hum Rights Watch 2021.
- [8] Fragkos P, Laura van Soest H, Schaeffer R, Reedman L, Köberle AC, Macaluso N, et al. Energy system transitions and low-carbon pathways in Australia, Brazil, Canada, China, EU-28, India, Indonesia, Japan, Republic of Korea, Russia and the United States. Energy 2021;216. 10.1016/j.energy.2020.119385.
- [9] Dolganova I, Rödl A, Bach V, Kaltschmitt M, Finkbeiner M. A review of life cycle assessment studies of electric vehicles with a focus on resource use. Resources 2020;9:1–26. https://doi.org/10.3390/resources9030032.
- [10] IEA. Net Zero by 2050: A Roadmap for the Global Energy Sector. 2021.
- [11] Sanguesa JA, Torres-Sanz V, Garrido P, Martinez FJ, Marquez-Barja JM. A review on electric vehicles: technologies and challenges. Smart Cities 2021;4:372–404. 10.3390/smartcities4010022.
- [12] Hao H, Liu Z, Zhao F, Geng Y, Sarkis J. Material flow analysis of lithium in China. Resour Policy 2017;51:100–6. https://doi.org/10.1016/j.resourpol.2016.12.005.
- [13] Gottesfeld P. Commentary health risks from climate fix: the downside of energy storage batteries. Environ Res 2019;178:108677. https://doi.org/10.1016/j. envres.2019.108677.
- [14] Hannula I, Reiner DM. Near-term potential of biofuels, electrofuels, and battery electric vehicles in decarbonizing road transport. Joule 2019;3:2390–402. https://doi.org/10.1016/j.joule.2019.08.013.
- [15] Del Pero F, Berzi L, Antonacci A, Delogu M. Automotive lightweight design: simulation modeling of mass-related consumption for electric vehicles. Machines 2020;8. https://doi.org/10.3390/machines8030051.
- [16] Miao Y, Hynan P, Von Jouanne A, Yokochi A. Current li-ion battery technologies in electric vehicles and opportunities for advancements. Energies 2019;12:1–20. https://doi.org/10.3390/en12061074.
- [17] Berzi L, Delichristov D, Favilli T, Pierini M, Ponchant M, Qehajaj A, et al. Smart Energy Management of Auxiliary Load for Electric Vehicles. Proc - 2020 IEEE Int Conf Environ Electr Eng 2020 IEEE Ind Commer Power Syst Eur EEEIC/I CPS Eur 2020 2020. 10.1109/EEEIC/ICPSEurope49358.2020.9160762.
- [18] Sovacool BK. The precarious political economy of cobalt: Balancing prosperity, poverty, and brutality in artisanal and industrial mining in the Democratic Republic of the Congo. Extr Ind Soc 2019;6:915–39. https://doi.org/10.1016/j. exis.2019.05.018.
- [19] Tabelin CB, Dallas J, Casanova S, Pelech T, Bournival G, Saydam S, et al. Towards a low-carbon society: a review of lithium resource availability, challenges and innovations in mining, extraction and recycling, and future perspectives. Miner Eng 2021;163:106743. https://doi.org/10.1016/j.mineng.2020.106743.
- [20] Sovacool BK, Hook A, Martiskainen M, Brock A, Turnheim B. The decarbonisation divide: contextualizing landscapes of low-carbon exploitation and toxicity in Africa. Glob Environ Chang 2020;60:102028. https://doi.org/10.1016/j. gloenvcha.2019.102028.
- [21] Findlay A. Dark side of low carbon. Nat Clim Chang 2020;10:184–184. 10.1038/ s41558-020-0724-1.
- [22] Sovacool BK, Hess DJ, Cantoni R. Energy transitions from the cradle to the grave: a meta-theoretical framework integrating responsible innovation, social practices, and energy justice. Energy Res Soc Sci 2021;75:102027. https://doi.org/ 10.1016/j.erss.2021.102027.
- [23] Sovacool BK, Heffron RJ, Mccauley D, Goldthau A. Energy decisions reframed as justice and ethical concerns 2016. 10.1038/NENERGY.2016.24.
- [24] Jenkins K, Mccauley D, Heffron R, Stephan H. Energy justice, a whole systems approach. Queens Polit Rev 2014;2:74–87.
- [25] Jenkins K, McCauley D, Heffron R, Stephan H, Rehner R. Energy justice: a conceptual review. Energy Res Soc Sci 2016;11:174–82. https://doi.org/ 10.1016/j.erss.2015.10.004.
- [26] Villavicencio Calzadilla P, Mauger R. The UN's new sustainable development agenda and renewable energy: the challenge to reach SDG7 while achieving energy justice. J Energy Nat Resour Law 2018;36:233–54. https://doi.org/ 10.1080/02646811.2017.1377951.
- [27] Mejía-Montero A, Lane M, van Der Horst D, Jenkins KEH. Grounding the energy justice lifecycle framework: an exploration of utility-scale wind power in Oaxaca. Mexico Energy Res Soc Sci 2021;75:102017. https://doi.org/10.1016/j. erss.2021.102017.

- [28] Cross J, Murray D. The afterlives of solar power: Waste and repair off the grid in Kenya. Energy Res Soc Sci 2018;44:100–9. https://doi.org/10.1016/j. erss.2018.04.034.
- [29] Sovacool BK, Lipson MM, Chard R. Temporality, vulnerability, and energy justice in household low carbon innovations. Energy Policy 2019;128:495–504. https:// doi.org/10.1016/j.enpol.2019.01.010.
- [30] Lacey-Barnacle M, Robison R, Foulds C. Energy justice in the developing world: a review of theoretical frameworks, key research themes and policy implications. Energy Sustain Dev 2020;55:122–38. https://doi.org/10.1016/j. esd.2020.01.010.
- [31] Sovacool BK, Martiskainen M, Hook A, Baker L. Decarbonization and its discontents: a critical energy justice perspective on four low-carbon transitions. 2019; 155. https://doi.org/10.1007/s10584-019-02521-7.
- [32] Sovacool BK, Hook A, Martiskainen M, Baker L. The whole systems energy injustice of four European low-carbon transitions. Glob Environ Chang 2019;58: 101958. https://doi.org/10.1016/j.gloenvcha.2019.101958.
- [33] IEA. Electric Vehicles. Paris: 2021.
- [34] Heffron RJ, McCauley D. The concept of energy justice across the disciplines. Energy Policy 2017;105:658–67. https://doi.org/10.1016/j.enpol.2017.03.018.
 [35] McCauley D, Heffron RJ, Stephan H, Jenkins K. Advancing energy justice: the
- triumvirate of tenets and systems thinking. Int Energy Law Rev 2013;32:107–10. [36] Jenkins KEH, Sovacool BK, Mouter N, Hacking N, Burns M-K, McCauley D. The
- (b) Sentilis Kerl, Sovacob BK, Mouter K, Itacking K, Burls WK, We karley D. The methodologies, geographies, and technologies of energy justice: a systematic and comprehensive review. Environ Res Lett 2021;16:043009. https://doi.org/ 10.1088/1748-9326/abd78c.
- [37] Hearn AX, Sohre A, Burger P. Innovative but unjust? Analysing the opportunities and justice issues within positive energy districts in Europe. Energy Res Soc Sci 2021;78:102127. https://doi.org/10.1016/j.erss.2021.102127.
- [38] Crootof A, Shrestha R, Albrecht T, Ptak T, Scott CA. Sacrificing the local to support the national: politics, sustainability, and governance in Nepal's hydropower paradox. Energy Res Soc Sci 2021;80:102206. https://doi.org/ 10.1016/j.erss.2021.102206.
- [39] Walker G. Beyond distribution and proximity: exploring the multiple spatialities of environmental justice. Antipode 2009;41:614–36.
- [40] Heffron RJ, McCauley D, Sovacool BK. Resolving society's energy trilemma through the Energy Justice Metric. Energy Policy 2015;87:168–76. https://doi. org/10.1016/j.enpol.2015.08.033.
- [41] Sovacool BK, Kester J, Noel L, de Rubens GZ. Energy injustice and nordic electric mobility: inequality, elitism, and externalities in the electrification of vehicle-togrid (V2G) transport. Ecol Econ 2019;157:205–17. https://doi.org/10.1016/j. ecolecon.2018.11.013.
- [42] Heffron R, Halbrügge S, Körner MF, Obeng-Darko NA, Sumarno T, Wagner J, et al. Justice in solar energy development. Sol Energy 2021;218:68–75. https:// doi.org/10.1016/j.solener.2021.01.072.
- [43] Heffron R, Körner MFM-F, Wagner J, Weibelzahl M, Fridgen G. Industrial demand-side flexibility: a key element of a just energy transition and industrial development. Appl Energy 2020;269:115026. https://doi.org/10.1016/j. appenrgy.2020.115026.
- [44] Jenkins K, McCauley D, Forman A. Energy justice: a policy approach. Energy Policy 2017;105:631–4. https://doi.org/10.1016/j.enpol.2017.01.052.
- [45] Barton J, Davies L, Dooley B, Foxon TJ, Galloway S, Hammond GP, et al. Transition pathways for a UK low-carbon electricity system: comparing scenarios and technology implications. Renew Sustain Energy Rev 2018;82:2779–90. https://doi.org/10.1016/j.rser.2017.10.007.
- [46] Kotak Y, Fernández CM, Casals LC, Kotak BS, Koch D, Geisbauer C, et al. End of electric vehicle batteries: reuse vs. recycle. Energies 2021;14:1–15. https://doi. org/10.3390/en14082217.
- [47] IEA. Global EV Outlook. 2020.
- [48] Kallitsis E, Korre A, Kelsall G, Kupfersberger M, Nie Z. Environmental life cycle assessment of the production in China of lithium-ion batteries with nickel-cobaltmanganese cathodes utilising novel electrode chemistries. J Clean Prod 2020;254: 120067. https://doi.org/10.1016/j.jclepro.2020.120067.
- [49] Lebedeva N, Persio DF, Boon-Brett L. Lithium ion battery value chain and related opportunities for Europe. vol. EUR 28534. 2017. 10.2760/6060.
- [50] Wen W, Yang S, Zhou P, Gao SZ. Impacts of COVID-19 on the electric vehicle industry: evidence from China. Renew Sustain Energy Rev 2021;144:111024. https://doi.org/10.1016/j.rser.2021.111024.
- [51] European Commission. Critical Raw Materials Resilience: Charting a Path towards greater Security and Sustainability. Brussels: 2020.
- [52] US Geological Survey. Mineral Commodity Summaries. 2021.
 [53] Amarakoon S, Smith J, Segal B. Application of Life- Cycle Assessment to
- Nanoscale Technology: Lithium-ion Batteries for Electric Vehicles. 2013. [54] Grupo de Trabalho. Relatório do Grupo de Trabalho "Lítio." Lisboa: 2017.
- [55] Zhao Y, Pohl O, Bhatt AI, Collis GE, Mahon PJ, Rüther T, et al. A review on battery market trends, second-life reuse, and recycling. Sustain Chem 2021;2: 167–205. https://doi.org/10.3390/suschem2010011.
- [56] ATIC. The Lithium-Ion Battery Value Chain New Economy Opportunities for Australia. Aust Trade Invest Comm 2018:56.
- [57] Harrison D, Ludwig C. Electric Vehicle Battery Supply Analysis: How Battery Demand and Production Are Reshaping the Automotive Industry. n.d.
- [58] CIC energiGUNE. World Map of Gigafactories 2021.
- [59] Automotive from Ultima Media. Electric vehicle and hybrid vehicle plant database 2021.
- [60] Bonsu NO. Towards a circular and low-carbon economy: insights from the transitioning to electric vehicles and net zero economy. J Clean Prod 2020;256: 120659. https://doi.org/10.1016/j.jclepro.2020.120659.

- [61] Qiao Q, Zhao F, Liu Z, Hao H. Electric vehicle recycling in China: Economic and environmental benefits. Resour Conserv Recycl 2019;140:45–53. https://doi.org/ 10.1016/j.resconrec.2018.09.003.
- [62] Meng F, McNeice J, Zadeh SS, Ghahreman A. Review of lithium production and recovery from minerals, brines, and lithium-ion batteries. Miner Process Extr Metall Rev 2021;42:123–41. https://doi.org/10.1080/08827508.2019.1668387.
- [63] Harper G, Sommerville R, Kendrick E, Driscoll L, Slater P, Stolkin R, et al. Recycling lithium-ion batteries from electric vehicles. Nature 2019;575:75–86. https://doi.org/10.1038/s41586-019-1682-5.
- [64] Ambrose H, O'Dea J. Electric vehicle batteries. Union Concerned Sci 2021:1–9. 10.1149/2.f05961if.
- [65] International Institute for Sustainable Development. Sustainability and Second Life: The case for cobalt and lithium recycling 2019:1–68.
- [66] Reinhardt R, Christodoulou I, Gassó-Domingo S, Amante GB. Towards sustainable business models for electric vehicle battery second use: a critical review. J Environ Manage 2019;245:432–46. https://doi.org/10.1016/j. ienvman.2019.05.095.
- [67] Kaunda RB. Potential environmental impacts of lithium mining. J Energy Nat Resour Law 2020;38:237–44. https://doi.org/10.1080/ 02646811.2020.1754596.
- [68] Farjana SH, Huda N, Mahmud MAP. Life cycle assessment of cobalt extraction process. J Sustain Min 2019;18:150–61. https://doi.org/10.1016/j. jsm.2019.03.002.
- [69] Egbue O. Assessment of social impacts of lithium for electric vehicle batteries. 62nd IIE Annu Conf Expo 2012 2012:1314–20.
- [70] Durán AP, Rauch J, Gaston KJ. Global spatial coincidence between protected areas and metal mining activities. Biol Conserv 2013;160:272–8. https://doi.org/ 10.1016/j.biocon.2013.02.003.
- [71] Agusdinata DB, Liu W, Eakin H, Romero H. Socio-environmental impacts of lithium mineral extraction: towards a research agenda. Environ Res Lett 2018;13. https://doi.org/10.1088/1748-9326/aae9b1.
- [72] Flexer V, Baspineiro CF, Galli CI. Lithium recovery from brines: a vital raw material for green energies with a potential environmental impact in its mining and processing. Sci Total Environ 2018;639:1188–204. https://doi.org/10.1016/ j.scitotenv.2018.05.223.
- [73] Hancock L, Ralph N, Ali SH. Bolivia's lithium frontier: Can public private partnerships deliver a minerals boom for sustainable development? J Clean Prod 2018;178:551–60. https://doi.org/10.1016/j.jclepro.2017.12.264.
- [74] Penn I, Lipton E. The Lithium Gold Rush: Inside the Race to Power Electric Vehicles. New York Times 2021.
- [75] Ribeiro T, Lima A, Vasconcelos C. The need for transparent communication in mining: a case study in lithium exploitation. Int J Sci Educ Part B Commun Public Engagem 2021;11:324–43. https://doi.org/10.1080/21548455.2021.1999530.
- [76] Chaves C, Pereira E, Ferreira P, Guerner Dias A. Concerns about lithium extraction: a review and application for Portugal. Extr Ind Soc 2021;8. 10.1016/j. exis.2021.100928.
- [77] Armand M, Axmann P, Bresser D, Copley M, Edström K, Ekberg C, et al. Lithiumion batteries – Current state of the art and anticipated developments. J Power Sources 2020;479. https://doi.org/10.1016/j.jpowsour.2020.228708.
- [78] Piçarra A, Annesley IR, Otsuki A, de Waard R. Market assessment of cobalt: identification and evaluation of supply risk patterns. Resour Policy 2021;73: 102206. https://doi.org/10.1016/j.resourpol.2021.102206.
- [79] Gulley AL, McCullough EA, Shedd KB. China's domestic and foreign influence in the global cobalt supply chain. Resour Policy 2019;62:317–23. https://doi.org/ 10.1016/j.resourpol.2019.03.015.
- [80] Hao H, Mu Z, Jiang S, Liu Z, Zhao F. GHG Emissions from the production of lithium-ion batteries for electric vehicles in China. Sustain 2017;9. 10.3390/ su9040504.
- [81] Sovacool BK. Who are the victims of low-carbon transitions? Towards a political ecology of climate change mitigation. Energy Res Soc Sci 2021;73:101916. https://doi.org/10.1016/j.erss.2021.101916.
- [82] Di Felice LJ, Renner A, Giampietro M. Why should the EU implement electric vehicles? Viewing the relationship between evidence and dominant policy solutions through the lens of complexity. Environ Sci Policy 2021;123:1–10. https://doi.org/10.1016/j.envsci.2021.05.002.
- [83] Dartford K, AFP. Locals and environment groups in David and Goliath battle over Tesla's Berlin "Gigafactory" 2021.
- [84] Günther HO, Kannegiesser M, Autenrieb N. The role of electric vehicles for supply chain sustainability in the automotive industry. J Clean Prod 2015;90:220–33. https://doi.org/10.1016/j.jclepro.2014.11.058.
- [85] Onat NC, Kucukvar M, Tatari O. Conventional, hybrid, plug-in hybrid or electric vehicles? State-based comparative carbon and energy footprint analysis in the United States. Appl Energy 2015;150:36–49. https://doi.org/10.1016/j. appenergy.2015.04.001.
- [86] Lelieveld J, Klingmüller K, Pozzer A, Burnett RT, Haines A, Ramanathan V. Effects of fossil fuel and total anthropogenic emission removal on public health and climate. Proc Natl Acad Sci USA 2019;116:7192–7. https://doi.org/10.1073/ pnas.1819989116.
- [87] Sironval V, Reylandt L, Chaurand P, Ibouraadaten S, Palmai-Pallag M, Yakoub Y, et al. Respiratory hazard of Li-ion battery components: Elective toxicity of lithium cobalt oxide (LiCoO₂) particles in a mouse bioassay. Arch Toxicol 2018;92: 1673–84. https://doi.org/10.1007/s00204-018-2188-x.
- [88] Shen M, Ren M, Wang Y, Shen F, Du R, Quan L, et al. Identifying dust as the dominant source of exposure to heavy metals for residents around battery factories in the Battery Industrial Capital of China. Sci Total Environ 2021;765: 144375. https://doi.org/10.1016/j.scitotenv.2020.144375.

- [89] Xiong S, Ji J, Ma X. Comparative life cycle energy and GHG emission analysis for BEVs and PHEVs: a case study in China. Energies 2019;12:1–17. https://doi.org/ 10.3390/en12050834.
- [90] CEPII. BACI database. BACI HS6 REV 1992 (1995 2020) 2021. http://www.ce pii.fr/CEPII/en/bdd_modele/bdd_modele_item.asp?id=37 (accessed May 14, 2022).
- [91] Liu Z, Song J, Kubal J, Susarla N, Knehr KW, Islam E, et al. Comparing total cost of ownership of battery electric vehicles and internal combustion engine vehicles. Energy Policy 2021;158:112564. https://doi.org/10.1016/j.enpol.2021.112564.
- [92] UN Environment Programme. Used vehicles and the Environment A global overview of used light duty vehicles: Flow, Scale and Regulation. 2020.
- [93] Ayetor GK, Mbonigaba I, Sackey MN, Andoh PY. Vehicle regulations in Africa: impact on used vehicle import and new vehicle sales. Transp Res Interdiscip Perspect 2021;10:100384. https://doi.org/10.1016/j.trip.2021.100384.
- [94] Doumbia M, Toure NE, Silue S, Yoboue V, Diedhiou A, Hauhouot C. Emissions from the road traffic of West African cities: assessment of vehicle fleet and fuel consumption. Energies 2018;11:1–16. https://doi.org/10.3390/en11092300.
- [95] André M, Pasquier A, Carteret M. Experimental determination of the geographical variations in vehicle fleet composition and consequences for assessing lowemission zones. Transp Res Part D Transp Environ 2018;65:750–60. https://doi. org/10.1016/j.trd.2018.10.005.
- [96] Morales Betancourt R, Galvis B, Rincón-Riveros JM, Rincón-Caro MA, Rodriguez-Valencia A, Sarmiento OL. Personal exposure to air pollutants in a bus rapid transit system: impact of fleet age and emission standard. Atmos Environ 2019; 202:117–27. https://doi.org/10.1016/j.atmosenv.2019.01.026.
- [97] Dillman KJ, Árnadóttir Á, Heinonen J, Czepkiewicz M, Davíðsdóttir B. Review and meta-analysis of EVs: embodied emissions and environmental breakeven. Sustain 2020;12:1–32. https://doi.org/10.3390/su12229390.
- [98] Moro A, Lonza L. Electricity carbon intensity in European Member States: impacts on GHG emissions of electric vehicles. Transp Res Part D Transp Environ 2018;64: 5–14. https://doi.org/10.1016/j.trd.2017.07.012.
- [99] Shaukat N, Khan B, Ali SM, Mehmood CA, Khan J, Farid U, et al. A survey on electric vehicle transportation within smart grid system. Renew Sustain Energy Rev 2018;81:1329–49. https://doi.org/10.1016/j.rser.2017.05.092.
- [100] Dranka GG, Ferreira P. Electric vehicles and biofuels synergies in the brazilian energy system. Energies 2020;13. https://doi.org/10.3390/en13174423.
- [101] Nunes P, Brito MC. Displacing natural gas with electric vehicles for grid stabilization. Energy 2017;141:87–96. https://doi.org/10.1016/j. energy.2017.09.064.
- [102] Zhou Y, Cao S. Energy flexibility investigation of advanced grid-responsive energy control strategies with the static battery and electric vehicles: a case study of a high-rise office building in Hong Kong. Energy Convers Manag 2019;199:111888. https://doi.org/10.1016/j.enconman.2019.111888.
- [103] Mrozik W, Rajaeifar MA, Heidrich O, Christensen P. Environmental impacts, pollution sources and pathways of spent lithium-ion batteries. Energy Environ Sci 2021;14:6099–121. https://doi.org/10.1039/D1EE00691F.

- [104] Connor P, Wise P. An Analysis of Lithium-ion Battery Fires in Waste Management and Recycling. 2021.
- [105] European Commission. Regulation of the European Parliament and of the Council concerning batteries and waste batteries, repealing Directive 2006/66/EC and amending Regulation (EU) No 2019/1020. 2020.
- [106] Chung D, Elgqvist E, Santhanagopalan S. Automotive Lithium-ion Cell Manufacturing: Regional Cost Structures and Supply Chain Considerations. 2016.
- [107] IEA. Electricity by generation source. Electr Inf 2020.[108] Hawkins TR, Singh B, Majeau-Bettez G, Strømman AH. Comparative
- [109] Hawkins H, Singin S, Majeau-Bettez G, Summan An. Comparative environmental life cycle assessment of conventional and electric vehicles. J Ind Ecol 2013;17:53–64. https://doi.org/10.1111/j.1530-9290.2012.00532.x.
 [109] IEA. Global EV Outlook 2021 - Accelerating ambitions despite the pandemic. Glob
- EV Outlook 2021 2021:101. [110] Lowans C, Furszyfer Del Rio D, Sovacool BK, Rooney D, Foley AM. What is the
- [110] Lowans C, Furszyler Der Rio D, Sovacool BK, Robiey D, Poley AM. what is the state of the art in energy and transport poverty metrics? A critical and comprehensive review. Energy Econ 2021;101:105360. https://doi.org/10.1016/ j.eneco.2021.105360.
- [111] Ghosh A. Possibilities and challenges for the inclusion of the electric vehicle (EV) to reduce the carbon footprint in the transport sector: a review. Energies 2020;13. https://doi.org/10.3390/en13102602.
- [112] McCauley D, Ramasar V, Heffron RJ, Sovacool BK, Mebratu D, Mundaca L. Energy justice in the transition to low carbon energy systems: exploring key themes in interdisciplinary research. Appl Energy 2019;233–234:916–21. https://doi.org/ 10.1016/j.apenergy.2018.10.005.
- [113] Dillman KJ, Czepkiewicz M, Heinonen J, Davíðsdóttir B. A safe and just space for urban mobility: a framework for sector-based sustainable consumption corridor development. Glob Sustain 2021;4. https://doi.org/10.1017/sus.2021.28.
- [114] Normann S. Green colonialism in the Nordic context: exploring Southern Saami representations of wind energy development. J Community Psychol 2021;49: 77–94. https://doi.org/10.1002/jcop.22422.
- [115] Annandale M, Meadows J, Erskine P. Indigenous forest livelihoods and bauxite mining: a case-study from northern Australia. J Environ Manage 2021;294: 113014. https://doi.org/10.1016/j.jenvman.2021.113014.
- [116] Skeete JP, Wells P, Dong X, Heidrich O, Harper G. Beyond the EVent horizon: battery waste, recycling, and sustainability in the United Kingdom electric vehicle transition. Energy Res Soc Sci 2020;69:101581. https://doi.org/10.1016/j. erss.2020.101581.
- [117] Malode SJ, Prabhu KK, Mascarenhas RJ, Shetti NP, Aminabhavi TM. Recent advances and viability in biofuel production. Energy Convers Manag X 2021;10: 100070. https://doi.org/10.1016/j.ecmx.2020.100070.
- [118] Hassani A, Maleki A. Projection of passenger cars' fuel demand and greenhouse gas emissions in Iran by 2050. Energy Convers Manag X 2021;12:100126. https:// doi.org/10.1016/j.ecmx.2021.100126.